Author’s response to reviews

Title: Profiles of lipids, blood pressure and weight changes among premenopausal Chinese breast cancer patients after adjuvant chemotherapy

Authors:

Winnie Yeo (winnie@clo.cuhk.edu.hk)
Frankie Mo (frankie@clo.cuhk.edu.hk)
Elizabeth Pang (liz@cuhk.edu.hk)
Joyce Suen (joyce@cuhk.edu.hk)
Jane Koh (jane@cuhk.edu.hk)
Herbert Loong (herbert@clo.cuhk.edu.hk)
Christopher Yip (chris@cuhk.edu.hk)
Rita Mg (rita@cuhk.edu.hk)
Claudia Yip (claudia@cuhk.edu.hk)
Nelson Tang (nelson@cuhk.edu.hk)
Giok Liem (giok@cuhk.edu.hk)

Version: 2 Date: 05 Jun 2017

Author’s response to reviews:

Responses to Comments from Reviewer 1:
Nil required.

Responses to Comments from Reviewer 2:
We have amended the description of the study according to the recommendation of the Reviewer. Please refer to Abstract, under Background, as well as Background section, paragraph 3.
Responses to Editor’s Comments:

The total number of patients who were approached for the study and the number who declined participation have now been described in the manuscript. Please refer to Results, paragraph 1.